Author’s response to reviews

Title: Prevalence of Undiagnosed Type 2 diabetes in patients admitted with acute coronary syndrome: The utility of easily reproducible screening methods

Authors:

Muhammad Ali Karamat (m.a.karamat@bham.ac.uk)
Umar Raja (umar_yousaf@hotmail.com)
Susan Manley (susan.manley@uhb.nhs.uk)
Alan Jones (alan.jones@heartofengland.nhs.uk)
Wasim Hanif (wasim.hanif@uhb.nhs.uk)
Abd Tahrani (abd.tahrani@nhs.net)

Version: 1 Date: 28 Dec 2016

Author’s response to reviews:

Editor comments

These have been addressed and the declaration section has got all the relevant information

Reviewer #1

Were any of the patients that were followed up 3 months later provided any medical treatment of their dysglycemia (metformin, etc.)? Yes those with initial HbA1c>7.5% were started on treatment and did not undergo a second OGTT (n=8).This information is included in the methods section under the heading OGTT. One patient with levels above 10% was treated with insulin.

Reviewer 2

Thanks for your comments.

1. One cannot start a sentence with a number which is not spelled out..

Line 30, page 1, abstract. 43% should be written as forty-three percent,

Line 3, Page 9, 198 patients without diabetes were approached.
These have been corrected in the manuscript

2. HbA1c has been written as HBA1C at many places especially in the results section.

This has been corrected in the manuscript

We have now corrected both the points raised